Author’s response to reviews

Title: Genomic profiles and transcriptomic microenvironments in 2 patients with synchronous lung adenocarcinoma and lung squamous cell carcinoma: a case report

Authors:

Licheng Wu (jam_woo@hotmail.com)
Poming Kang (kangbert@msn.com)
Shaolin Tao (taoshaolin2004@163.com)
Zhikun Zhao (zhaozhikun@yucebio.com)
Longyun Chen (chenly@yucebio.com)
Yajie Xiao (xiaoyajie@yucebio.com)
Qunyou Tan (dpxwk2019@163.com)

Version: 1 Date: 29 Oct 2019

Author’s response to reviews:

Dear editor,

The main corrections are as following:

1. In the "Consent for publication" section, please clarify if Consent for publication was received from both patients.
   Response: we have modified the "Consent for publication" part to "The written informed consents for publication were obtained from both patients".

2. At this stage, we ask that you submit a clean version of your manuscript and do not include track changes or highlighting.
   Response: we have removed all the track changes and highlighting in the manuscript.

Thank you for your comments!

Best,
Qunyou Tan